Author’s response to reviews

Title: Using the Distance between Sets of Hierarchical Taxonomic Clinical Concepts to Measure Patient Similarity

Authors:

Zheng Jia (zjia@zju.edu.cn)
Xudong Lu (lvxu@zju.edu.cn)
Huilong Duan (duanhl@zju.edu.cn)
Haomin Li (hmli@zju.edu.cn)

Version: 4 Date: 27 Mar 2019

Author’s response to reviews:

1. Please upload the additional files Data Interpretation and Subpopulation Criteria.docx and Supplemental results.docx.

Re: We have uploaded the additional files.

2. Please remove the "response to editor and reviewer.docx" from the additional files as it should not form part of the final publication.

Re: The response to editor and reviewer.docx have been removed from the additional files.

3. Please include a statement on ethics approval and consent to participate. All manuscripts describing studies involving human participants, human data, animal data or human tissue require a statement on ethics approval and consent (even where the need for approval was waived). Please provide the full name of the ethics committee that approved the study and the committee’s reference number if appropriate.

Re: The Ethics approval and consent to participate Section was updated with detail information.

4. Please state in the Methods section whether written informed consent for participation in the study was obtained from participants or, where participants are children, a parent or guardian.
RE: A section named "data set for this study" was added to the Methods section. And the ethics approval and the waive of patient consent were stated.